Author’s response to reviews

Title: Risk factors associated with abscess formation among patient with leg erysipelas (cellulitis) in sub-Saharan Africa: a multicenter study.

Authors:

Palokinam Vincent Pitche (vincent.pitche@gmail.com)
Bayaki Saka (barthelemysaka@yahoo.fr)
Ahy Boubacar Diatta (ahydiatta@yahoo.com)
Ousmane Faye (faye_o@yahoo.fr)
Boh Fanta Diané (bohfanta2010@yahoo.fr)
Abdoulaye Sangaré (sang_abdoulaye@yahoo.fr)
Pascal Niamba (niamba_pascal@yahoo.com)
Christine Mandengue (cmmandengue@udesmontagnes.org)
Léonon Kobengue (kobangleon@yahoo.fr)
Assane Diop (kobangueleon@gmail.com)
Fatinata Ly (lyfaty@yahoo.fr)
Mame Thierno Dieng (mafath@orange.sn)
Alassane Dicko (ppitche@yahoo.fr)
Mohamed Macire Soumah (medsoum7@yahoo.fr)
Mohamed Cissé (cissebibi1@gmail.com)
Sarah Hamdan Kourouma (sarambilal@yahoo.fr)
Isidore Kouassi (isidore.kouassi@yahoo.fr)
Taniratou Boukari (taniratou_boukari@yahoo.fr)
Sefako Akakpo (sefasyl@yahoo.fr)
Dadja Essoya Landoh (mlandoh@yahoo.fr)
Author’s response to reviews:

Dear Editor-in-Chief,
BMC Dermatology journal,

Re: BDER-D-15-00002
Risk factors associated with abscess formation among patient with leg erysipelas (superficial cellulitis) in sub-Saharan Africa: a multicenter study.

I am pleased to resubmit for publication the revised version of our manuscript BDER-D-15 00002 “Risk factors associated with abscess formation among patient with leg erysipelas (superficial cellulitis) in sub-Saharan Africa: a multicenter study”.

The manuscript has been revised by Dr NAQI Ali, a Dermatologist and scientific writer. He has reviewed and copyedited the manuscript.

We very much hope we have provided satisfactory corrections and hope that the manuscript is now acceptable for publication in your Journal.

Yours sincerely

Doctor Dadja Essoya Landoh
OMS-Togo
BP: 1396 Lomé (Togo).
E-mail: mlandoh@yahoo.fr